Author’s response to reviews

Title: Notch1 signaling induces Epithelial-Mesenchymal Transition in lens epithelium cells during hypoxia

Authors:

Lei Liu (leilei_315@hotmail.com)
Wei Xiao (estherllsy@hotmail.com)

Version: 4 Date: 19 Jul 2017

Author’s response to reviews:

We modified the manuscript according to your advices.

1. Reformulate the manuscript especially the Methods section.

2. Ethics approval and consent to participate

The local ethics committee ruled that no formal ethics approval was required in this particular case.

3. Availability of data and material

All data generated or analysed during this study are included in this published article. More details are available from the corresponding author on reasonable request.

4. Competing interests

The authors declare that they have no competing interests.

5. Authors’ Contributions

 LL performed the experiments, analyzed the data and wrote the manuscript. WX supervised the experiments and did the critical review and final approval of the manuscript. All authors read and approved the final manuscript.